



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
Fax: +1-925-223-8243
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

ANSWERING REVIEWERS

Name of journal: World Journal of Diabetes
Manuscript NO: 47209
Title: Management of Diabetic Dyslipidemia: An Update
Reviewer's code: 02631746
Reviewer's country: United States
Science editor: Ying Dou
Reviewer accepted review: 2019-03-12 02:04
Reviewer performed review: 2019-03-13 19:27
Review time: 1 Day and 17 Hours

SPECIFIC COMMENTS TO AUTHORS

Good review. It will be much better if the authors can insert a table or figure outlining the actions of various drugs used in the management of dyslipidemia in DM and their mechanisms of action for easy reading/understanding.

We thank the reviewer for the positive comments. The table is added to the revised paper.

THE MANUSCRIPT

Google Search:

- The same title
- Duplicate publication
- Plagiarism
- No



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
Fax: +1-925-223-8243
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

BPG Search:

- The same title
- Duplicate publication
- Plagiarism
- No